RESEARCH

Open Access

Check for updates

Maternal immune suppression during pregnancy does not prevent abnormal behavior in offspring

Ashley Griffin¹, Teylor Bowles², Lucia Solis³, Teryn Railey², Samer Beauti⁴, Reanna Robinson³, Shauna-Kay Spencer², James P Shaffery⁵ and Kedra Wallace^{2,3*}¹

Abstract

Background Offspring of hypertensive disorders of pregnancy are at an increased risk of developing neurodevelopmental and neurobehavioral disorders compared to offspring from non-affected pregnancies. Using rodent models of Preeclampsia (PreE; new onset of hypertension after 20 weeks gestation) and HELLP (hemolysis, elevated liver enzymes, and low platelets), we studied the behavioral outcome of their offspring in adolescence.

Methods A subset of dams received Orencia, a T-cell activation inhibitor, as T cells have been associated with the induction of hypertension and inflammation during pregnancy. We hypothesized that offspring from hypertensive dams would experience adverse behavioral outcomes in social, cognitive, locomotor, and anxiety tests, and offspring from dams treated with Orencia would demonstrate less adverse behaviors.

Results Male offspring of PreE + Orencia dams (p < 0.05) and female offspring from HELLP + Orencia dams (p < 0.05) spent more time playing compared to normal pregnant offspring. All offspring from hypertensive and Orencia-treated dams performed worse on the Barnes Maze test compared to normal pregnant. We also measured adult (postnatal day > 60) myelin basic protein (MBP) and NeuN expression in both the prefrontal cortex and hippocampus. In the hippocampus and prefrontal cortex, there was no difference in expression of either MBP or NeuN in all groups regardless of sex.

Conclusion The results from this study suggest that offspring of hypertensive disorders of pregnancy have behavioral changes, specifically cognitive differences. This study has shown that there is a sex dependent difference in offspring neurobehavioral development, influenced in part by the type of hypertensive disorder of pregnancy, and alterations in the maternal immune system.

Highlights

- Female offspring from HELLP dams treated with or without Orencia (Abatacept, prevents T cell activation) have decreased locomotion.
- Male offspring from PreE dams and females from HELLP dams that have been treated with Orencia have increased social play.

*Correspondence: Kedra Wallace kwallace2@umc.edu

Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicate of the original autory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Deciration waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

- Offspring of both hypertensive and Orencia treated dams have spatial memory deficits.
- Male offspring from PreE dams treated with Orencia show evidence of anxiety-like behavior in marble burying tasks.

Plain language summary

Children of pregnancies that are complicated by hypertensive disorders of pregnancy (HDP) such as Preeclampsia and HELLP (hemolysis, elevated liver enzymes, and low platelets) syndrome have an increased risk of having behavioral deficits and changes in brain development. Both Preeclampsia and HELLP have been shown to activate the immune and inflammation systems in the body of the mother. In this study, we used offspring of rat models of Preeclampsia and HELLP to study their behavior including anxiety-like behaviors and memory deficits. We also compared offspring of rat models of Preeclampsia and HELLP that were given Orencia, which minimizes immune responses by blocking the activation of T cells. We also studied two regions of the brain (prefrontal cortex and hippocampus) to measure two proteins (myelin basic protein (MBP) and NeuN) involved in brain function. Our study found that offspring from dams that were treated with Orencia during pregnancy with HDP had sex differences in time playing. All offspring, regardless of the HDP dam being treated with or without Orencia, had evidence of spatial learning deficits. When sexes and groups were compared there was no difference in MBP or NeuN expression in the prefrontal cortex or hippocampus.

Keywords sFlt-1, Preeclampsia, HELLP syndrome, ADHD, Neurodevelopment, Hypertensive disorders of pregnancy, Offspring behavior, Sex differences

Introduction

Maternal cardiovascular disorders during pregnancy have long-term effects on offspring that can range from developmental to metabolic disorders [1]. Hypertensive disorders of pregnancy (HDP) such as Preeclampsia (PreE, new onset of both hypertension and proteinuria presenting after 20 weeks gestation) and HELLP (hemolysis, elevated liver enzymes, and low platelets) affect approximately 3-8% of pregnancies worldwide [2, 3]. The most commonly accepted pathophysiological mechanism for PreE and HELLP is that inadequate invasion of the uterine spiral arteries leads to abnormal oxygenation of the placenta followed by an imbalance of angiogenic factors [4]. Currently the treatment course for both disorders is for the woman to end the pregnancy through delivery of the placenta; which will stop the more acute effects of the disorders. The timing of this delivery most often results in a premature delivery of human infants, and subsequently offspring from women are often small for gestational age and preterm which increases the risk of neurodevelopmental disorders [5, 6].

Children born to women with HDP like PreE and HELLP have also been reported to have endocrine and cardiovascular morbidities as adolescents and adults [7–9]. There is also increased reports of attention-deficit/ hyperactivity disorder (ADHD), intellectual disabilities, autism spectrum disorder and cerebral palsy among children born to women with HDP [10–13]. One study found that as the maternal condition worsened from PreE to severe PreE/HELLP so did the hazard of childhood mental disorders, there is a link between severity of HDP and negative impact on the affected children [14].

Outside of the context of HDP, maternal immune activation during pregnancy also leads to behavioral abnormalities in children [15]. Experimental studies have reported that immune activation during pregnancy is associated with not only HDPs such as PreE and HELLP but also with behavioral abnormalities in offspring from postnatal week 1 up unto postnatal month 3 [16-19]. When the immune system is activated, T cells stimulate an inflammatory response. In both animal models of HELLP and PreE, it has been shown that there is an increase circulating inflammatory cytokines and CD4⁺ T cells, accompanied by hypertension [20-22]. In pregnant women it is well established that maternal immune activation through viral or bacterial infection leads to altered neurobehavioral outcomes, but it is unclear how maternal inflammation stemming from HDP affect the behavioral outcomes of the offspring [23, 24]. Immune activation, specifically an increase in the T cell derived inflammatory Th17 cells, is associated with the pathogenesis of PreE [25, 26]. We have previously reported that blockade of T cell activation with Orencia (Abatacept; a selective co-stimulation modulator which inhibits T cell activation) during pregnancy, prevents the inflammatory and hypertensive response in PreE and HELLP animal models [21, 22]. In the current study we investigated whether offspring born to HDP dams had neurobehavioral disturbances and if suppression of the maternal immune response during pregnancy could reduce or prevent any neurobehavioral disturbances.

Materials and methods

Induction of PreE and HELLP in dams and animal grouping All animals used in this study were offspring born to Sprague Dawley dams (originating from Envigo; Indianapolis, IN). Animals were housed in a temperature-controlled room with a reversed 12:12 light: dark cycle and fed standard rat chow. All procedures were approved by the Institutional Animal Care and Use Committee at the University of Mississippi Medical Center (UMMC) under protocol number 1414 A.

To induce experimental PreE (n=16 dams), miniosmotic pumps (model 2002, Alzet Scientific, Cupertino, CA) infusing soluble Fms like tyrosine kinase-1 (sFlt-1; 7 µg/kg, R&D Systems, Minneapolis, MN) was mixed with saline and inserted into the intraperitoneal cavity of pregnant dams beginning on gestational day (GD) 12 [27]. Experimental HELLP (n=13 dams) was induced via a similar method, however a combination of sFlt-1 and soluble Endoglin (sEng; 7 and 4.7 µg/kg respectively, R&D Systems) was infused via mini-osmotic pumps [20]. Normal pregnant (NP) dams (n=24) had an abdominal incision made with no mini-osmotic pump insertion. This group represented an uncomplicated pregnancy and served as a normotensive control group. As previous studies have shown that infusion of saline does not alter the cardiovascular, metabolic or immune response in NP dams, pumps were not inserted and only the SHAM surgery was conducted. To attenuate maternal immune activation and hypertension during pregnancy, Orencia (2 mg/kg) was infused into a subset of NP (n=10), PreE (n=11) and HELLP (n=9) dams on GD13 via the jugular vein as previously reported [21].

All dams delivered between GD21-22. Immediately after delivery of pups, up to eight pups per litter were removed and cross-fostered with non-study lactating dams with no more than 2 pups from the same litter being placed within the same new litter. The different litters of pups were identified via color markings by using a Sharpie to mark the animals on their back and tail, to identify animals. This mechanism of identification was chosen as there is no pain or a risk of infection compared to ear tagging or tattooing. The initial color marking was performed by an independent investigator to ensure experimenters were blind to pups maternal group. All animals were weaned from foster dams between postnatal days 30-33. At this time they remained in mixed treatment groups and were placed in cages with samesex cage mates. The following pups were used in the current study: NP male (n=21), NP female (n=21), HELLP male (n=15), HELLP female (n=14), PreE male (n=17), PreE female (n=16), NP+O male (n=15), NP+O female (n=8), HELLP+O male (n=12), HELLP+O female (n=10), PreE+O male (n=11), PreE+O female (n=7).

Behavior

All animal testing was conducted in the Animal Behavior Core at UMMC and was completed during the animal's active phase and under low light conditions. Animals were given an hour to acclimate to the testing room unless otherwise noted. For tests that required tracking, the Noldus Observer (Noldus Information Technology, Inc, Leesburg, VA) was used to record and track behavioral testing and a video recorder was used to record sessions that required scoring or validation at a later time period.

Social interactions and distance traveled

On postnatal day (PND) 24–26, social interaction and self-initiated play behavior (chasing, following, boxing and pinning) and self-isolation behaviors (self-grooming, self-exploring and time at rest) were measured to assess social play and hyperactivity. Rats were placed in the test arena (60 cm x 30 cm x 40 cm) with a non-familiar conspecific of the same sex and allowed to interact with one another for 20 min. All test sessions were recorded by a camera to allow subsequent scoring which was conducted as previously described [28]. On PND 38–39, open field test was used to assess locomotor activity. Rats were allowed to explore the empty chamber (40 cm x 50 cm x 50 cm) for 20 min in which the distance traveled was captured.

Barnes maze

Beginning on PND45, rats were trained in Barnes Maze to assess spatial memory [29]. Rats were placed in a maze surrounded by a set of distinct stationary black and white social cues for 5 days. Rats were placed daily into the goal box (i.e., escape chamber) for a four-minute adaptation period. After four minutes, rats were returned to a hold-ing cage and the goal box was cleaned with 10% ethanol. Rats were then placed directly in the center of the plat-form and allowed to explore the maze for 5 min. If the goal box was found before time elapsed, that time was automatically recorded and they were allowed to remain in the chamber for 60 s before they were returned to the home cage. If the escape chamber was not located, the subject was returned to the home cage at the end of the trial and the time marked as 301 s.

Marble burying

On PND59 anxiety-like behavior was assessed via marble burying. 15 glass marbles were evenly distributed across an empty cage lined with fresh bedding. Rats were allowed 15 min to interact with the marbles before being returned to their home cages and the number of buried marbles was counted. Marbles were considered buried if at least two-thirds were covered by sawdust, with the more marbles that were buried correlating with increased anxiety-like behavior [30]. Cages and marbles were cleaned with 70% ethanol and the sawdust was replaced between subjects.

Mean arterial pressure (MAP) measurement and organ collection

On PND60 all pups underwent surgery and an indwelling catheter made of V1, V3 and V6 tubing (Scientific Commodities, Lake Havasu City, AZ, USA) was inserted into the carotid artery and then exteriorized and secured at the back of the neck. Rats were allowed to recover overnight and on the following day all rats had their MAP assessed. Briefly, rats were placed in individual restrainers and carotid lines connected to pressure transducers (Cobe III Transducer CDX Sema, Birmingham AL, USA). Following a 30-min acclimation period, MAP was measured for 30-min. After which while under anesthesia, whole blood was collected as were brains. For future studies, the kidney, liver, and spleen were weighed and saved at -80 °C.

Western blot

After euthanization brains were weighed, dissected and stored at -80 °C. The prefrontal cortex and hippocampus regions were homogenized in Tris lysis buffer containing protease inhibitors, centrifuged and the protein concentration was determined using a bicinchoninic acid protein assay (BCA; ThermoFisher, Rockford, IL) as previously described [31]. Briefly, samples were electrophoresed and transferred onto nitrocellulose membranes using Bio-Rad (Bio-Rad Laboratories, Hercules, CA) gels, membranes, and transfer system, followed by confirmation of transfer via Ponceau-S (Sigma Aldrich, St. Louis, MO) staining. Membranes were blocked with 5% milk buffer, washed in TBST (tris buffered saline with 0.1% tween 20) and allowed to incubate overnight at 4 °C with either mouse anti-neuronal nuclear (NeuN; 1:1000; Catalog #MA5-33103, Bio-legend, San Diego, CA) or Mouse anti-myelin basic protein (MBP; 1:1000; Catalog #808,401, Bio-legend). After washing in TBST, membranes were incubated with 1:1000 anti-mouse HRPconjugated secondary antibody (Catalog #R1005, Kindle Biosciences, Greenwich, CT), followed by development with KwikQuant Digital enhanced chemiluminescence (Kindle Biosciences). Membranes were imaged with the KwikQuant imager imaging system and protein bands analyzed using Image J software. NeuN and MBP expression were normalized to the amount of total protein present within the sample lane as detected by Ponceau-S [32].

Statistics

A two-way ANOVA was used to compare differences between groups and sex differences within groups. The ANOVAs were followed by Tukey's multiple comparisons test or student t-tests were used for additional analysis as needed. For the Barnes Maze analysis, repeated measure two-way ANOVA was performed. Statistical analyses were performed using GraphPad Prism 10. All data were reported as mean \pm standard error mean. Values of p < 0.05 were considered statistically significant.

Results

Maternal Orencia administration increases social play behavior in male offspring from PreE dams and females from HELLP dams

We examined the effect of HDP with or without immune suppression via Orencia administration on social behavior interactions. The analysis of social play showed a significant group effect ($F_{2.60} = 6.63$, p=0.003), treatment effect ($F_{1.60} = 4.24$, p = 0.04) and a significant interaction between groups, treatment and sex ($F_{2.60} = 5.08$, p=0.009, Fig. 1A). Post-hoc analysis showed that males from PreE+Orencia dams spent significantly more time socially playing compared to males from NP dams (p=0.02) and female offspring from NP (p=0.003), NP+Orencia (p=0.03) and HELLP (p=0.03) dams. Female offspring from HELLP+Orencia dams spent more time socially playing relative to female offspring from NP dams (p=0.02). When isolation time (the time spent alone) was analyzed there was not a significant main effect on group ($F_{2.60} = 1.32$, p=0.28), however there were significant effects on both treatment $(F_{1.60})$ = 5.51, p=0.02) and sex (F_{1.60} = 4.87, p=0.03, Fig. 1B). Post-hoc analysis only indicated that female offspring $(928.5\pm99.9 \text{ s})$ from NP dams spent significantly more time alone vs. male offspring from PreE+Orencia dams $(585.4 \pm 151.5 \text{ s}; p=0.02).$

Female offspring from HELLP and HELLP + Orencia dams have reduced locomotion activity

There was not a significant effect on sex ($F_{1,80} = 1.69$, p=0.2) or treatment ($F_{1,80} = 2.1$, p=0.15) in locomotor activity assessed by distance traveled. There was a significant main effect on group ($F_{2,80} = 3.7$, p=0.03, Fig. 1D) and a significant treatment by sex effect ($F_{1,80} = 12.44$, p=0.0007). Post-hoc analysis indicated that female off-spring from HELLP dams traveled significantly less relative to NP offspring (p=0.02 in males and p=0.008 in females). Whereas Orencia treatment to moms further reduced distance traveled in female offspring when compared to male offspring from NP+Orencia (p=0.04) and HELLP+Orencia (p=0.005) dams and female offspring from PreE+Orencia dams (p=0.003).

Offspring from HDP dams and Orencia treated dams show evidence of impairments in spatial learning and memory

Because sex did not have a significant effect on performance in the Barnes Maze (Data Supplement), sexes



Fig. 1 Pups born to dams who had administration of Orencia to otherwise HDP of pregnancy dams showed difference in social play behavior, isolation, and locomotion. Social play behavior separated by sex (**A**), isolation time separated by sex (**B**) and distance travelled separated by sex (**C**) Abbreviations: NP (normal pregnant); PreE (preeclampsia); HELLP (hemolysis elevated liver enzyme low platelet). The open squares denote males and open circles denote females. 1 A legend: *a* denotes p < 0.05 between the indicated group and male offspring from NP dams; female offspring from NP, NP + Orencia, and HELLP dams. *b* denotes p < 0.05 between the indicated group and female offspring from NP dams. 1B legend: *c* denotes p < 0.05 between indicated group and female offspring and NP male and female offspring. *e* denotes p < 0.05 between indicated group and female PreE + Orencia offspring. *e* denotes p < 0.05 between indicated group and female PreE + Orencia offspring.

were collapsed for the subsequent analyses. There was not a main effect on group ($F_{5.89} = 1.9$, p=0.10) in the Barnes Maze, however significant main effects were present when the days ($F_{4.356} = 15.92$, p < 0.0001), subjects $(F_{89, 356} = 2.07, p < 0.0001)$ and the interaction between groups and days ($F_{20.356} = 1.69$, p = 0.03) were examined (Fig. 2A). On day 1 of testing offspring from NP+Orencia dams took significantly longer (p < 0.05) to complete the task compared to every group except for offspring from PreE+Orencia dams, whereas by day 5 of the test offspring from NP dams completed the task significantly faster relative to offspring from all other groups (p=0.002). Post-hoc analysis found that offspring from NP (p < 0.0001), HELLP (p = 0.01), NP+Orencia (p = 0.03) and PreE+Orencia (p=0.03) dams demonstrated significant improvement over the testing period relative to Day 1.

Offspring from PreE, PreE+ Orencia and NP + Orencia dams have increased anxiety

When anxiety was assessed via marble burying there were no significant effects on sex ($F_{1,85} = 2.48$, p=0.12) but there was a significant main effect between groups ($F_{5,85} = 2.92$, p=0.02) where offspring from NP dams buried significantly less marbles (i.e. less anxiety) relative to offspring from PreE (p=0.01), NP+Orencia (p=0.003) and PreE+Orencia dams (p=0.02, Fig. 2B). There was a significant interaction between sex and group ($F_{2,85} = 3.59$, p=0.03) where male offspring from PreE+Orencia dams buried significantly more marbles relative to males from NP (p=0.02) and HELLP (p=0.04) dams and when



Fig. 2 Offspring of HDP regardless of Orencia treatment have spatial learning deficits and anxiety-like behavior. Spatial learning was observed over 5 days in the Barnes Maze (**A**) and anxiety was assessed via the number of marbles buried over a 15-minute period (**B**). There were no significant changes in mean arterial pressure between males and females (**C**). 2 A legend: a denotes that on day 1 of testing offspring from NP + Orencia dams took significantly longer to complete the tasks vs. all other groups except PreE + Orencia. *b* denotes that on day 5 of testing NP offspring completed the task significantly faster relative to all other groups. 2B legend: *c* denotes p < 0.05 between the indicated group and NP male, NP female and HELLP male offspring

compared to female offspring from NP dams (p=0.04). There was also not a significant effect of maternal Orencia treatment offspring marble burying ($F_{1,91}$ = 1.44, p=0.23).

No changes in blood pressure or protein expression

MAP was assessed to determine if offspring from hypertensive dams were also susceptible to hypertension. There were no significant differences between groups due to sex ($F_{1,106} = 0, p=1$), group ($F_{2,106} = 0.19, p=0.82$) or Orencia treatment ($F_{2,106} = 1.84$, p = 0.18; Fig. 2C). Finally, neural expression of MBP and NeuN were evaluated as both have a role in the neural connections associated with learning and memory. There was no significant main effect on sex in either MBP prefrontal cortex ($F_{1,27} = 1.02$, p=0.32) or hippocampus (F_{1,34} = 1.21, p=0.28) protein expression or with NeuN prefrontal cortex ($F_{1,30} = 0.83$, p=0.37) or hippocampus (F_{1.30} = 3.55, p=0.07) expression, therefore sex data was collapsed within groups. In the prefrontal cortex, there was a significant difference between groups in MBP ($F_{2.33} = 4.29, p = 0.02$, Fig. 3A, E) with post-hoc analysis indicating no specific group differences. There were no significant differences in NeuN protein expression ($F_{2,36} = 0.93$, p=0.40, Fig. 3B, F). Similar results were seen in the hippocampal expression of MBP

($F_{2,40} = 0.16$, p=0.85, Fig. 3C, G) and NeuN ($F_{2,36} = 0.5$, p=0.61, Fig. 3D, H). In no cases were there any significant differences in protein expression due to treatment of Orencia (Data Supplement).

Discussion

The current study evaluated if offspring born to hypertensive dams treated with or without an immune suppressant were susceptible to behavioral abnormalities. PreE and/or HELLP have been reported to have over activation of the immune system which is believed to contribute to the pathophysiology of these pregnancy disorders [33, 34]. Clinical studies have identified an association among children born to women with HDP in which the children present with mood disorders (anxiety and depression), autism, cognitive deficits and ADHD [10, 35, 36]. To determine if suppression of immune activation improves behavior among adolescent offspring, pregnant dams were treated with Orencia in an effort to prevent CD4⁺ T cell activation. Important to this growing body of work are the new data that we provide demonstrating sex specific differences in offspring from both PreE and HELLP dams and further insight into the effects of Orencia administration during pregnancy.



Fig. 3 No difference in MBP and NeuN expression in the hippocampus and prefrontal cortex between groups. Representative western blot of homogenates (30 μg) from the prefrontal cortex (**A**, **E**) and (**B**, **F**) and the hippocampus region (**C**, **G**) and (**D**, **H**) with expression of Myelin Basic Protein and NeuN respectively depicting the characteristic bands at 20 kDa for MBP and 45–55 kDa for NeuN, and bar graphs showing the relationship between groups and sex within each respective brain region. Data are represented as mean ±SEM; *n*=3–5/group/sex

Social interactions and play behavior in rats during the adolescent period are measurements of sociocognitive skills that are relatable to the development of social behavior in humans [37, 38], therefore we wanted to evaluate play behavior in the current study. Offspring from PreE and HELLP dams responded differently in the behavioral challenges, indicating that the degree of maternal hypertension and immune activation in utero impacts the behavior of offspring. Immune suppression in HDP led to evidence of increased social play. Overall, these offspring showed increased time socially interacting and playing, which did not directly translate to an increase in distance traveled. Again supporting the idea of hyperactivity or enhanced social play among some groups. Opposite effects were seen in offspring from normotensive rats subjected to immune suppression; these rats preferred to spend more time in isolation while there was not a great effect on their distance traveled. There was also an increase in marble burying activity among male PreE+Orencia offspring. The increased number of buried marbles has been interpreted to be both an indicator of increased anxiety-like behavior as well as the repetitive behavior that is seen in individuals along the autism spectrum or with obsessive-compulsive disorders [39]. To determine if these behavioral results are more in line with the repetitive behavior seen in neurodevelopmental disorders vs. anxiety-like behavior, additional behavioral studies of these disorders would need to be conducted.

Preclinical studies have shown behavioral and sex differences in offspring of maternal immune activation models where male offspring were more likely to exhibit anxiety and deficits in spatial working memory relative to female offspring [40, 41]. Impairments in spatial learning and memory have also been reported by others, as offspring from the L-NAME model of PreE have impairments in spatial learning and memory [42, 43]. Immune suppression in the normotensive dams led to offspring having worse outcomes on day 1 of the spatial learning test. By the end of the test, offspring fromPreE dam had no significant improvement of spatial learning from baseline. In addition, regardless of hypertension or immune suppression all offspring had learning impairments compared to normotensive offspring. Immune suppression decreased anxiety-like behavior in offspring from HDP rats. The immune suppression during pregnancy has a similar effect on the maternal presentation of anxiety-like behavior [27, 44].

Abnormal myelination in both clinical and pre-clinical studies has been correlated with cognitive deficits in frontal-temporal and hippocampal regions of the brain in psychotic disorders, ADHD, and Alzheimer's [45–47]. Studies have also found that increased MBP and NeuN expression among offspring exposed to *in utero* inflammation was found to be associated with improvements in learning and memory [48, 49]. Despite these reports, MBP, responsible for myelinating nerves, and NeuN were not increased in either the prefrontal cortex or the hippocampus in the current study. It has been reported that the peak time for MBP expression in rodents is PND 20, after which expression stabilizes [50], if this is the case it might explain the lack of differences among our groups. Interestingly, PreE offspring (from the L-NAME model) that expressed impairments in spatial learning and memory, also had changes in cortical, hippocampal and striatal NeuN expression when assessed via immunohistochemistry [42, 51]. The findings of no change in blood pressure among offspring from HDP dams in the current study is similar to another study with similar findings when offspring from PreE dams were examined [52].

There are several different animal models of maternal immune activation and of HDP, many of which utilize different methods to induce hypertension and/or immune activation. Despite the difference in methodologies, our results are similar to what has been reported among male offspring. In an arginine vasopressin model of PreE, male offspring of PreE dams exhibited increased social behavior in the three-chamber social preference task relative to control offspring, without there being a significant difference in distance travelled between the groups [53]. Our study is not without limitations, firstly even though it has previously been demonstrated that these animal models have hypertension and inflammation during pregnancy we did not directly evaluate that in the dams that birthed the offspring used in the current study [20, 54]. We also did not measure the immune profile of these offspring as immune activation during pregnancy can be correlated with neurodevelopmental outcome of offspring [55]. Future studies need to include a variety of test to assess learning and memory to further elucidate the different roles that the prefrontal cortex and hippocampus have these offspring and to determine if there are sex differences.

Clinically Orencia is only administered during pregnancy when the maternal benefit outweighs the potential risk to the fetus. However, despite this classification clinical studies have found no association between Orencia exposure in utero and fetal death or the development of congenital anomalies [56]. The findings from the current study indicate that despite the previous reports of maternal benefit during pregnancy [21, 22], maternal immune suppression as a result of Orencia may have negative effects on offspring. This is especially true in instances where there was likely not a robust immune response such as in NP rats. Immune suppression in the absence of an immune response or via CD4⁺ regulatory T cell depletion has been linked with increased anxiety-like behavior, depression and alterations in social behavior [57, 58] and might in fact be one reason why marble burying is increased in our control NP+Orencia offspring.

Perhaps due in part to the lower incidence of HELLP, the majority of clinical studies evaluating the long-term effects of HDP on offspring do not distinguish between mothers diagnosed with PreE vs. HELLP. There were also distinct differences between male and female off-spring both in their behavior and in the degree and rate of brain maturation, emphasizing the increased need for additional studies to delineate sex differences. The results from the current study add to the growing body of evidence indicating that HELLP is a different maternal indication rather than a subtype of PreE [59, 60], and highlight the importance of evaluating the classifications of HDP when assessing neurodevelopment [61].

Perspectives and significance

This study was able to demonstrate that Orencia administration during a preeclamptic and HELLP pregnancy led to different effects on social interactions and locomotion. These different effects also were sex dependent. Overall Orencia did not improve behavior in offspring from HDP rats, it is possible that the timing of immune suppression during pregnancy had detrimental effects on the offspring (as seen in NP+Orencia offspring) rather than protective effects. Previous studies have also reported that when Orencia is administered during pregnancy and animals are evaluated during pregnancy, babies from HELLP+Orencia dams are smaller relative to untreated dams [21]. Whether the offspring was from an HDP dam or Orencia treated dam, there was evidence of learning impairments independent of sex. This study adds to the growing body of knowledge indicating that PreE and HELLP Syndrome may have a differential effect on offspring behavioral development which is further complicated by sex and possible in utero exposure to immune therapeutics.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s13293-024-00600-8.

Supplementary Material 1

Supplementary Material 2

Acknowledgements

Not applicable.

Author contributions

KW contributed to the conception and design of the study. TB, LS, RR and AG contributed to the behavioural data collection while JPS contributed to the behavioural methodology. SB and TR contributed to data collection. AG and TR analysed the data and SKS supervised the study. AG wrote the first draft of the manuscript. All authors contributed to manuscript revision, read and approved the submitted version of the manuscript.

Funding

Research supported in this publication was supported in part by the National Institutes of Health (NIH) under Award Numbers R01MH116027 to KW, T32HL105324 and William Townsend Porter predoctoral fellowship from the American Physiological Society to A.G. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

Data availability

All data generated or analyzed during the study are included in this published article and its supplementary information files.

Declarations

Ethics approval and consent to participate

All procedures were approved by the Institutional Animal Care and Use Committee at the University of Mississippi Medical Center (UMMC) under protocol number 1414 A.

Consent for publication

Not applicable.

Competing Interests

The authors declare that they have no competing interests.

Author details

¹Program in Neuroscience, University of Mississippi Medical Center, 2500 North State Street, Jackson, MS 39216, USA

 ²Department of Pharmacology & Toxicology, University of Mississippi Medical Center, 2500 North State Street, Jackson, MS 39216, USA
³Department of Obstetrics & Gynecology, University of Mississippi Medical Center, 2500 North State Street, Jackson, MS 39216, USA
⁴Master's in Biomedical Science program, School of Graduate Studies in Health Sciences, University of Mississippi Medical Center, 2500 North State Street, Jackson, MS 39216, USA

⁵Department of Psychiatry and Human Behavior, University of Mississippi Medical Center, 2500 North State Street, Jackson, MS 39216, USA

Received: 9 October 2023 / Accepted: 4 March 2024 Published online: 26 March 2024

References

- 1. Wiener S, Wolfe D. Links between maternal cardiovascular disease and the health of offspring. Can J Cardiol. 2021;37:12.
- Chappell L, Cluver C, Kingdom J, Tong S. Pre-eclampsia. Lancet. 2021;398:10297341–54.
- Khan B, Allah Yar R, Khakwani AK, Karim S, Arslan Ali H. Preeclampsia Incidence and its maternal and neonatal outcomes with Associated Risk factors. Cureus. 2022;14:11e31143. https://doi.org/10.7759/cureus.31143.
- Rana S, Lemoine E, Granger JP, Karumanchi SA, Preeclampsia. Pathophysiology, challenges, and perspectives. Circ Res. 2019;124:71094–112. https://doi. org/10.1161/circresaha.118.313276.
- Macdonald-Wallis C, Tilling K, Fraser A, Nelson S, Lawlor D. Associations of blood pressure change in pregnancy with fetal growth and gestational age at delivery: findings from a prospective cohort. Hypertension. 2014;64:36–44.
- An H, Jin M, Li Z, Zhang L, Li H, Zhang Y, et al. Impact of gestational hypertension and pre-eclampsia on preterm birth in China: a large prospective cohort study. BMJ Open. 2022;12:9e058068. https://doi.org/10.1136/ bmjopen-2021-058068.
- Sacks K, Frigr M, Shoham-Vardi I, Spiegel E, Sergienko R, Landau D, Sheiner E. Prenatal exposure to preeclampsia as an independent risk factor for long-term cardiovascular morbidity of the offspring. Pregnancy Hypertens. 2018;13:181–6.
- Tenhola S, Rahiala E, Martikainen A, Halonen P, Voutilainen R. Blood pressure, serum lipids, fasting insulin and adrenal hormones in 12-year old children born with maternal preeclampsia. J Clin Endocrinol Metab. 2003;88:3.
- Huang C, Wei K, Lee PMY, Qin G, Yu Y, Li J. Maternal hypertensive disorder of pregnancy and mortality in offspring from birth to young adulthood: national population based cohort study. BMJ. 2022;379:e072157. https://doi. org/10.1136/bmj-2022-072157.
- Maher G, Dalman C, O'Keeffe G, Kearney P, McCarthy F, Kenny L, Khashan A. Association between preeclampsia and attention-deficity hyperactivity

disorder: a population-based and sibling-matched cohort study. Acta Psychiatr Scand. 2020;142:4275–83.

- 11. Dachew B, Mamun A, Maravilla J, Alati R. Pre-eclampsia and the risk of autismspectrum disorder in offspring: meta-analysis. Br J Psychiatry. 2018;212:3.
- Wang H, Laszlo K, Gissler M, Li F, Zhang J, Yu Y, Li J. Maternal hypertensive disorders and neurodevelopmental disorders in offspring: a population-based cohort in two nordic countries. Eur J Epidemiol. 2021;36:5519–30.
- Sun BZ, Moster D, Harmon QE, Wilcox AJ. Association of Preeclampsia in Term Births with Neurodevelopmental disorders in offspring. JAMA Psychiatry. 2020;77:8.
- Lahti-Pulkkinen M, Girchenko P, Tuovinen S, Sammallahti S, Reynolds R, Lahti J, et al. Maternal hypertensive pregnancy disorders and mental disorders in children. Hypertension. 2020;75:6.
- Quagliato LA, de Matos U, Nardi AE. Maternal immune activation generates anxiety in offspring: a translational meta-analysis. Transl Psychiatry. 2021;11:1. https://doi.org/10.1038/s41398-021-01361-3.
- Ponzio N, Servatius R, Beck K, Marzouk A, Kreider T. Cytokine levels during pregnancy influence immunological profiles and neurobahvioral patterns of the offspring. Ann NY Acad Sci. 2007;1107:118–28.
- Griffin A, Spencer S-K, Bowles T, Solis L, Robinson R, Ramarao S, Wallace K. Male HELLP pups experience sensorimotor delays and reduced body weight. Physiol Behav. 2021;241:113567.
- Ballendine S, Greba Q, Dawicki W, Zhang X, Gordon J, Howland J. Behavioral alterations in rat offspring following maternal immune activation and ELR-CXC chemokine receptor antagonism during pregnancy: implications for neurodevelopmental psychiatric disorders. Prog Neuropsychopharmacol Biol Psychiatry. 2015;57:155–65.
- LaMarca B, Speed J, Fournier L, Babcock S, Berry H, Cockrell K, Granger J. Hypertension in response to chronic reductions in uterine perfusion in pregnant rats: effect of tumor necrosis factor-alpha blockade. Hypertension. 2008;52:6.
- Wallace K, Morris R, Kyle PB, Cornelius D, Darby M, Scott J, et al. Hypertension, inflammation and T lymphocytes are increased in a rat model of HELLP syndrome. Hypertens Pregnancy. 2014;33:1:41–54. https://doi.org/10.3109/10 641955.2013.835820.
- Bean C, Spencer S, Bowles T, Kyle P, Williams J, Gibbens J, Wallace K. Inhibition of T cell-activation attenuates hypertension, TNF-alpha, IL-17 and bloodbrain barrier permeability in pregnant rats with angiogenic imbalance. Am J Reprod Immunol. 2016;76:4.
- 22. Novotny S, Wallace K, Herse F, Moseley J, Darby M, Heath J, et al. CD4+T cells play a critical role in mediating hypertension in response to placental ischemia. J Hypertens. 2013;2:2:1–6.
- Massrali A, Adhya D, Srivastava DP, Baron-Cohen S, Kotter MR. Virus-Induced maternal Immune activation as an environmental factor in the etiology of Autism and Schizophrenia. Front Neurosci. 2022;16:834058. https://doi. org/10.3389/fnins.2022.834058.
- Smith SE, Li J, Garbett K, Mirnics K, Patterson PH. Maternal immune activation alters fetal brain development through interleukin-6. J Neurosci. 2007;27:40:10695–702. https://doi.org/10.1523/jneurosci.2178-07.2007.
- Fu B, Tian Z, Wei H. Th17 cells in human recurrent pregnancy loss and preeclampsia. Cell Mol Immunol. 2014;11:564–70.
- Nesselhut T, Hinney B, Rath W, Balwanz A, Cillien N, Kuhn W. Changes in the basic immunologic parameters interleukin-2 receptor and CD8-antigen in preeclampsia. Z Geburtshilfe Perinatol. 1992;196:263–6.
- Wallace K, Bowles T, Griffin A, Robinson R, Solis L, Railey T, et al. Evidence of anxiety, Depression and Learning impairments following prenatal hypertension. Behav Sci (Basel). 2022;12:2. https://doi.org/10.3390/bs12020053.
- Auger A, Olesen K. Brain sex differences and the organization of juvenile social play behavior. J Neuroendocrinol. 2009;21:6519–25.
- 29. Barnes C. Memory deficits associated with senescence: a neurophysiological and behavioral study in the rat. J Comp Physiol Pyschol. 1979;93:1.
- Njung'e K, Handley S. Evaluation of marble-burying behavior as a model of anxiety. Pharmacol Biochem Behav. 1991;38:163–7.
- Bean C, Spencer S, Pabbidi M, Szczepanski J, Araji S, Dixon S, Wallace K. Peripheral anti-angiogenic imbalance during pregnancy impairs myogenic tone and increases cerebral edema in a rodent model of HELLP syndrome. Brain Sci. 2018;8:12E216.
- 32. Myers E, Sanders G, Ravi D, Matchar D, Havrilesky L, Samsa G et al. Evaluating the Potential Use of Modeling and Value-of-Information Analysis for Future Research Prioritization within the evidence-based practice center program. Durham, NC: Duke Evidence-based practice center2011. Report No.: 5 Contract No.: 11-ECH030-EF.

- Lamarca B, Cornelius D, Harmon A, Amaral L, Cunningham M Jr, Faulkner J, Wallace K. Identifying immune mechanisms mediating the hypertension during preeclampsia. Am J Physiol Regul Integr Comp Physiol. 2016;311(1):R1–9.
- Stojanovska V, Zenclussen AC. Innate and adaptive Immune responses in HELLP syndrome. Front Immunol. 2020;11:667. https://doi.org/10.3389/ fimmu.2020.00667.
- 35. Many A, Fattal A, Leitner Y, Kupferminc M, Harel S, Jaffa A. Neurodevelopmental and cognitive assessmet nof children born growth restricted to mothers with and without preeclampsia. Hypertensi Pregnancy. 2003;22:25–9.
- 36. Tuovinen S, Eriksson J, Kajantie E, Lahti J, Pesonen A, Heinonen K et al. Maternal hypertensive disorders of pregnancy and self-reported cognitive impairment of the offpsirng 70 years later: the Helsinki Birth Cohort Study. American Journal of Obstetrics and Gynecology. 2013;208:3:200.e1-e9.
- Baarendse PJ, Counotte DS, O'Donnell P, Vanderschuren LJ. Early social experience is critical for the development of cognitive control and dopamine modulation of prefrontal cortex function. Neuropsychopharmacology. 2013;38:8:1485–94. https://doi.org/10.1038/npp.2013.47.
- Nijhof SL, Vinkers CH, van Geelen SM, Duijff SN, Achterberg EJM, van der Net J, et al. Healthy play, better coping: the importance of play for the development of children in health and disease. Neurosci Biobehav Rev. 2018;95:421– 9. https://doi.org/10.1016/j.neubiorev.2018.09.024.
- de Brouwer G, Fick A, Harvey B, Wolmarans D. A critical inquiry into marbleburying as a preclinical screening paradigm of relevance for anxiety and obsessive-compulsive disorder: mapping the way forward. Cogn Affect Behav Neurosci. 2019;19(1):1–39.
- Gogos A, Sbisa A, Witkamp D, Buuse M. Sex differences in the effect of maternal immune activation on cognitive and psychosis-like behaviour in Long Evans rats. Eur J Neurosci. 2020;52:1.
- 41. Xuan I, Hampson D. Gender-dependent effects of maternal immune activation on the behavior of mouse offspring. PLoS ONE. 2014;9:8e104433.
- Liu X, Zhao W, Liu H, Kang Y, Ye C, Gu W, et al. Developmental and functional brain impairment in offspring from Preeclampsia-Like rats. Mol Neurobiol. 2016;53(2):1009–19. https://doi.org/10.1007/s12035-014-9060-7.
- Zhu H, Zhu W, Hu R, Wang H, Ma D, Li X. The effect of pre-eclampsia-like syndrome induced by L-NAME on learning and memory and hippocampal glucocorticoid receptor expression: a rat model. Hypertens Pregnancy. 2017;36(1):36–43. https://doi.org/10.1080/10641955.2016.1228957.
- 44. Wallace K, Bean C, Bowles T, Spencer SK, Randle W, Kyle PB, Shaffery J. Hypertension, anxiety, and blood-brain barrier permeability are increased in Postpartum severe Preeclampsia/Hemolysis, elevated liver enzymes, and low platelet count syndrome rats. Hypertension. 2018;72:4946–54. https://doi. org/10.1161/hypertensionaha.118.11770.
- Papuć E, Kurys-Denis E, Krupski W, Tatara M, Rejdak K. Can antibodies against glial derived antigens be early biomarkers of hippocampal demyelination and memory loss in Alzheimer's Disease? J Alzheimers Dis. 2015;48:1115–21. https://doi.org/10.3233/jad-150309.
- Onnink AM, Zwiers MP, Hoogman M, Mostert JC, Dammers J, Kan CC, et al. Deviant white matter structure in adults with attention-deficit/hyperactivity disorder points to aberrant myelination and affects neuropsychological performance. Prog Neuropsychopharmacol Biol Psychiatry. 2015;63:14–22. https://doi.org/10.1016/j.pnpbp.2015.04.008.
- Vanes LD, Mouchlianitis E, Barry E, Patel K, Wong K, Shergill SS. Cognitive correlates of abnormal myelination in psychosis. Sci Rep. 2019;9:1. https://doi. org/10.1038/s41598-019-41679-z.

- Makinodan M, Tatsumi K, Manabe T, Yamauchi T, Makinodan E, Matsuyoshi H, et al. Maternal immune activation in mice delays myelination and axonal development in the hippocampus of the offspring. J Neurosci Res. 2008;86:102190–200. https://doi.org/10.1002/jnr.21673.
- Chen JF, Liu K, Hu B, Li RR, Xin W, Chen H, et al. Enhancing myelin renewal reverses cognitive dysfunction in a murine model of Alzheimer's disease. Neuron. 2021;109:142292–e3075. https://doi.org/10.1016/j. neuron.2021.05.012.
- 50. Wiggins RC. Myelination: a critical stage in development. Neurotoxicology. 1986;7:2103–20.
- Nakamura N, Ushida T, Onoda A, Ueda K, Miura R, Suzuki T, et al. Altered offspring neurodevelopment in an L-NAME-induced preeclampsia rat model. Front Pediatr. 2023;11:1168173. https://doi.org/10.3389/fped.2023.1168173.
- Lu F, Bytautiene E, Tamayo E, Gamble P, Anderson GD, Hankins GD, et al. Gender-specific effect of overexpression of sFlt-1 in pregnant mice on fetal programming of blood pressure in the offspring later in life. Am J Obstet Gynecol. 2007;197:4. https://doi.org/10.1016/j.ajog.2007.06.064.
- Gumusoglu SB, Chilukuri ASS, Hing BWQ, Scroggins SM, Kundu S, Sandgren JA, et al. Altered offspring neurodevelopment in an arginine vasopressin preeclampsia model. Transl Psychiatry. 2021;11:179. https://doi.org/10.1038/ s41398-021-01205-0.
- Deer E, Jones J, Cornelius DC, Comley K, Herrock O, Campbell N, et al. Progesterone Induced blocking factor reduces hypertension and placental mitochondrial dysfunction in response to sFIt-1 during pregnancy. Cells. 2021;10:11. https://doi.org/10.3390/cells10112817.
- Conway F, Brown AS. Maternal Immune activation and related factors in the risk of offspring Psychiatric disorders. Front Psychiatry. 2019;10:430. https:// doi.org/10.3389/fpsyt.2019.00430.
- 56. Kumar M, Ray L, Vemuri S, Simon T. Pregnancy outcomes following exposure to abatacept during pregnancy. Semin Arthritis Rheum. 2015;45:3.
- 57. Munshi S, Parrilli V, Rosenkranz J. Peripheral anti-inflammatory cytokine interleukin-10 treatment mitigates interleukin-1B-induced anxiety and sickness behaviors in adult male rats. Behav Brain Res. 2019;372:112024.
- Kim S, Lee H, Lee G, Oh S, Shin M, Shim I, Bae H. CD4+CD25+regulatory T cell depletion modulates anxiety and depression-like behaviors in mice. PLoS ONE. 2012;7:7e42054.
- Haram K, Mortensen J, Nagy B. Genetic aspects of preeclampsia and the HELLP syndrome. J Pregnancy. 2014. https://doi.org/10.1155/2014/910751.
- Lisonkova S, Bone JN, Muraca GM, Razaz N, Wang LQ, Sabr Y, et al. Incidence and risk factors for severe preeclampsia, hemolysis, elevated liver enzymes, and low platelet count syndrome, and eclampsia at preterm and term gestation: a population-based study. Am J Obstet Gynecol. 2021;225:5. https://doi. org/10.1016/j.ajog.2021.04.261.
- Barron A, McCarthy CM, O'Keeffe GW. Preeclampsia and Neurodevelopmental outcomes: potential pathogenic roles for inflammation and oxidative stress? Mol Neurobiol. 2021;58:6:2734–56. https://doi.org/10.1007/ s12035-021-02290-4.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.